Author's response to reviews

Title: Understanding implementation processes of clinical pathways and clinical practice guidelines in pediatric contexts: A study protocol

Authors:

Shannon D Scott (shannon.scott@ualberta.ca)
Jeremy Grimshaw (jgrimshaw@ohri.ca)
Terry P Klassen (tklassen@mich.ca)
Alberto Nettel-Aguirre (alberto.nettel-aguirre@albertahealthservices.ca)
David W Johnson (david.johnson@albertahealthservices.ca)

Version: 2 Date: 25 November 2011

Author's response to reviews: see over
Dear Dr Mittman & Dr Eccles,

We have reviewed the editorial changes you proposed for our manuscript titled “Understanding implementation processes of clinical pathways and clinical practice guidelines in pediatric contexts: A study protocol”, and found them satisfactory. We did notice that in the document that you sent, there were some spacing issues in the reference section (starts around reference #21) ---these appear to be problems with the software as we were not able to ‘fix’ the problem. Otherwise, we agree with the changes you have made and are happy to have you proceed.

Sincerely,

Shannon Scott
AHFMR, Population Health Investigator
CIHR, New Investigator
Faculty of Nursing, University of Alberta